Author’s response to reviews

Title: Cryptic Leishmania infantum infection in Italian HIV infected patients

Authors:

Claudia Colomba (claudia.colomba@libero.it)
Laura Saporito (laura.sapo@tin.it)
Fabrizio Vitale (fstefano.reale@izssicilia.it)
Stefano Reale (stefano.reale@izssicilia.it)
Giustina Vitale (vitale@unipa.it)
Alessandra Casuccio (casuccio@unipa.it)
Manlio Tolomeo (manliotolomeo@hotmail.com)
Daniela Maranto (dany.maranto@libero.it)
Raffaella Rubino (raffaellarubino@libero.it)
Paola Di Carlo (dicarlo19@libero.it)
Lucina Titone (titonel@unipa.it)

Version: 9 Date: 25 November 2009

Author’s response to reviews:

Dear Sir,

Please find herewith included the revised version of the manuscript entitled:
Cryptic Leishmania infantum infection in Italian HIV infected patients.
The requested formatting changes have been made. Particularly, figure 1 and 2
have been cited in the results section.

Thank you.

Regards,

Claudia Colomba